Author’s response to reviews

Title: Endothelial Cell-Specific Molecule-1 as an Invasiveness Marker for Pituitary Null Cell Adenoma

Authors:

Shousen Wang (wshsen1965@126.com)
Zhifeng Wu (turanwangle@sina.com)
Liangfeng Wei (wlf74@163.com)
Jianhe Zhang (leo200303034@126.com)

Version: 3 Date: 07 Aug 2019

Author’s response to reviews:

Aug 7, 2019

Dr. Zuleyha Karaca
BMC Endocrine Disorders
https://bmcendo.endocrinology.biomedcentral.com/

Re: Ms. Ref. No.: BEND-D-19-00130R2

Dear Dr. Zuleyha Karaca,

We thank you for the opportunity to revise our manuscript, entitled “Endothelial Cell-Specific Molecule-1 as an Invasiveness Marker for Pituitary Null Cell Adenoma”.

Detailed point-by-point replies to the reviewers’ comments in addition to our revised manuscript have been resubmitted. All changes have been made in the revised manuscript. We hope our revised manuscript could be acceptable for publication in BMC Endocrine Disorders. We thank you once more for your time and consideration.

Sincerely,

Shousen Wang

Reviewer reports:
In the section 'Ethics Approval and Consent to Participate', please state whether the informed consent obtained was written or verbal. If verbal, please state the reason and whether the ethics committee approved this procedure.

Response: We’ve revised the sentence accordingly.

Please add a “Conclusions” section after the “Discussion” section. This should state clearly the main conclusions of the research article and give a clear explanation of their importance and relevance.

Response: We’ve added the conclusion section.